

# **JGR** Biogeosciences

# **RESEARCH ARTICLE**

10.1029/2021JG006543

#### **Key Points:**

- Abrupt permafrost thaw turned the tundra into a substantial annual source of CO<sub>2</sub> of which 25%–31% were released in the non-growing season
- About 0.8% of thawed permafrost carbon was decomposed to CO<sub>2</sub> in one year but decomposition rates declined after the loss of labile carbon
- Methane contributed a minor fraction to total greenhouse gas fluxes also because of a low methanogen abundance in Pleistocene permafrost

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

C. Knoblauch, Christian.Knoblauch@uni-hamburg.de

#### **Citation:**

Knoblauch, C., Beer, C., Schuett, A., Sauerland, L., Liebner, S., Steinhof, A., et al. (2021). Carbon dioxide and methane release following abrupt thaw of Pleistocene permafrost deposits in Arctic Siberia. *Journal of Geophysical Research: Biogeosciences*, *126*, e2021JG006543. https://doi. org/10.1029/2021JG006543

Received 23 JUL 2021 Accepted 13 OCT 2021

#### **Author Contributions:**

Conceptualization: Christian Beer, Alexander Schuett, Lewis Sauerland, Eva-Maria Pfeiffer Formal analysis: Christian Beer, Susanne Liebner, Axel Steinhof, Janet Rethemeyer, Alexey Faguet Funding acquisition: Christian Beer, Eva-Maria Pfeiffer Investigation: Christian Beer, Alexander Schuett, Lewis Sauerland,

#### © 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

# Carbon Dioxide and Methane Release Following Abrupt Thaw of Pleistocene Permafrost Deposits in Arctic Siberia

Christian Knoblauch<sup>1,2</sup>, Christian Beer<sup>1,2</sup>, Alexander Schuett<sup>1,2</sup>, Lewis Sauerland<sup>1,3</sup>, Susanne Liebner<sup>4,5</sup>, Axel Steinhof<sup>6</sup>, Janet Rethemeyer<sup>7</sup>, Mikhail N. Grigoriev<sup>8</sup>, Alexey Faguet<sup>9</sup>, and Eva-Maria Pfeiffer<sup>1,2</sup>

<sup>1</sup>Institute of Soil Science, Universität Hamburg, Hamburg, Germany, <sup>2</sup>Center for Earth System Research and Sustainability, Universität Hamburg, Hamburg, Germany, <sup>3</sup>Now at Institute for Environmental Science (ACES), University of Stockholm, Stockholm, Sweden, <sup>4</sup>GFZ German Research Centre for Geosciences, Section Geomicrobiology, Potsdam, Germany, <sup>5</sup>Institute of Biochemistry and Biology, University of Potsdam, Potsdam, Germany, <sup>6</sup>Max Planck Institute for Biogeochemistry, Jena, Germany, <sup>7</sup>Institute of Geology and Mineralogy, University of Cologne, Cologne, Germany, <sup>8</sup>Russian Academy of Sciences, Siberian Branch, Mel'nikov Permafrost Institute, Yakutsk, Russia, <sup>9</sup>Trofimuk Institute of Petroleum Geology and Geophysics, Novosibirsk, Russia

**Abstract** The decomposition of thawing permafrost organic matter (OM) to the greenhouse gases (GHG) carbon dioxide  $(CO_2)$  and methane forms a positive feedback to global climate change. Data on in situ GHG fluxes from thawing permafrost OM are scarce and OM degradability is largely unknown, causing high uncertainties in the permafrost-carbon climate feedback. We combined in situ CO<sub>2</sub> and methane flux measurements at an abrupt permafrost thaw feature with laboratory incubations and dynamic modeling to quantify annual CO, release from thawing permafrost OM, estimate its in situ degradability and evaluate the explanatory power of incubation experiments. In July 2016 and 2019, CO, fluxes ranged between 0.24 and 2.6 g  $CO_2$ -C m<sup>-2</sup> d<sup>-1</sup>. Methane fluxes were low, which coincided with the absence of active methanogens in the Pleistocene permafrost. CO, fluxes were lower three years after initial thaw after normalizing these fluxes to thawed carbon, indicating the depletion of labile carbon. Higher CO<sub>2</sub> fluxes from thawing Pleistocene permafrost than from Holocene permafrost indicate OM preservation for millennia and give evidence that microbial activity in the permafrost was not substantial. Short-term incubations overestimated in situ CO, fluxes but underestimated methane fluxes. Two independent models simulated median annual  $\rm CO_2$  fluxes of 160 and 184 g  $\rm CO_2$ -C m<sup>-2</sup> from the thaw slump, which include 25%-31% CO<sub>2</sub> emissions during winter. Annual CO<sub>2</sub> fluxes represent 0.8% of the carbon pool thawed in the surface soil. Our results demonstrate the potential of abrupt thaw processes to transform the tundra from carbon neutral into a substantial GHG source.

**Plain Language Summary** Thawing of permanently frozen soils (permafrost) in the northern hemisphere forms a threat to global climate since these soils contain large amounts of frozen organic carbon, which might be decomposed to the greenhouse gases (GHGs) carbon dioxide ( $CO_2$ ) and methane upon thaw. How fast these GHGs are produced is largely unknown, since field observations of greenhouse gas fluxes from thawing permafrost are too sparse. Consequently, simulations on the effect of thawing permafrost soils on future climate are highly uncertain. We measured  $CO_2$  and methane fluxes from soils affected by abrupt permafrost thaw in Siberia during two summer seasons. We used these field observations and long-term incubation data to calibrate two models that simulate the  $CO_2$  release over a whole year. We found that greenhouse gas fluxes were dominated by  $CO_2$  and that the minor importance of methane was due to the absence of methane producing microorganisms in the Pleistocene permafrost. The  $CO_2$  release in the first year accounted for 0.8% of thawed permafrost carbon but decomposition rates decreased after the depletion of the rapidly decomposable organic matter. Abrupt permafrost thaw turned the tundra into a substantial source of  $CO_2$ , of which 25%–31% was released in the non-growing season.

# 1. Introduction

Perennially frozen ground (permafrost) of the northern hemisphere contains about 800 Pg organic carbon (Hugelius et al., 2014), which has been protected from decomposition for hundreds to thousands of years



10.1029/2021JG006543

Susanne Liebner, Axel Steinhof, Mikhail N. Grigoriev Methodology: Christian Beer, Alexander Schuett, Lewis Sauerland, Susanne Liebner, Axel Steinhof, Janet Rethemever Project Administration: Mikhail N. Grigoriev, Eva-Maria Pfeiffer Resources: Mikhail N. Grigoriev Software: Christian Beer Validation: Christian Beer Visualization: Alexey Faguet Writing - review & editing: Christian Beer, Alexander Schuett, Lewis Sauerland, Susanne Liebner, Axel Steinhof, Janet Rethemeyer, Mikhail N. Grigoriev

due to freezing temperatures. Another 500 Pg organic carbon are stored in the seasonally unfrozen surface active layer of permafrost soils and perennially unfrozen sediments (Hugelius et al., 2014). The active layer receives a regular input of fresh organic matter (OM) from the surface vegetation, which is decomposed by microorganisms during the summer thaw period. The observed permafrost thaw, caused by rising atmospheric temperature unlocks formerly frozen OM, which may thereupon be decomposed to the greenhouse gases (GHGs) carbon dioxide ( $CO_2$ ) and methane. The GHG release from thawing permafrost is seen as one of the major tipping elements in the climate due to the related positive feedback mechanism (Beer, 2008; Heimann & Reichstein, 2008). Still it is one of the key uncertainties when simulating the future global carbon cycle and climate. Model simulations on the fate of soil carbon in the permafrost region differ vastly between a net carbon gain of about 50 Pg and a loss of 140 Pg carbon until 2100 (McGuire et al., 2018).

This large range of model predictions is related to fundamental uncertainties about the decomposability of permafrost OM. Current estimates on how fast permafrost OM may be mineralized to  $CO_2$  after thaw range between 15% in 100 years and 66% in one summer thaw season (Elberling et al., 2013; Knoblauch et al., 2013; Plaza et al., 2019; Schädel et al., 2014; Schuur et al., 2015; Vonk et al., 2012). While some biomarker and fractionation studies indicate its rapid decomposability (Mueller et al., 2015; Strauss et al., 2015), long-term laboratory incubations demonstrate the dominance of slowly degradable OM with turnover times of hundreds of years (Knoblauch et al., 2013; Schädel et al., 2014; Schuer et al., 2014). Results from laboratory incubations are subsequently used to constrain simulations of future GHG release from permafrost landscapes (Koven et al., 2015; Schneider von Deimling et al., 2015). However, laboratory incubations may not consider all of the environmental parameters that strongly affect *in situ* OM turnover, such as gradients of temperature, oxygen and moisture, freezing and thawing or the input of fresh OM (Pegoraro et al., 2019; Schädel et al., 2016; Walz et al., 2017; Wang et al., 2020; Wild et al., 2016). Thus, it still remains unclear how far organic carbon decomposition rates from laboratory studies represent those under *in situ* conditions.

Most studies on GHG fluxes from thawing permafrost landscapes report ecosystem respiration ( $R_{eco}$ ) fluxes from vegetated soils affected by thawing permafrost (Abbott & Jones, 2015; Fouche et al., 2014; Mauritz et al., 2017; Natali et al., 2015; Pegoraro et al., 2021; Ravn et al., 2020). These CO<sub>2</sub> fluxes originate from both microbial decomposition of soil OM [heterotrophic respiration ( $R_h$ )] and plant respiration [autotrophic respiration ( $R_a$ )]. Furthermore,  $R_h$  fluxes include both CO<sub>2</sub> from OM decomposition of recently fixed carbon in the surface active layer and from older permafrost carbon thawing at the soil's bottom (Mauritz et al., 2021). A direct measurement of *in situ* turnover of recently thawed permafrost OM is challenging since permafrost thaws in typically 0.3–1 m depth at the bottom of the active layer.

The few studies that estimated GHG fluxes from permafrost OM used a combination of field manipulations, incubation studies and <sup>14</sup>C analysis (Cooper et al., 2017; Estop-Aragonés et al., 2018; Hicks Pries et al., 2016; Schuur et al., 2009) but *in situ* measurements of permafrost OM decomposition are still rare. Since the degradability of deep permafrost OM may substantially differ from those of the surface active layer (Cooper et al., 2017; De Baets et al., 2016; Schädel et al., 2014; Song et al., 2020), knowledge on the degradability of permafrost OM under *in situ* conditions is needed to constrain future GHG fluxes from thawing permafrost landscapes.

The aims of the current study were to (a) quantify the *in situ* decomposability of Pleistocene and Holocene permafrost OM, (b) evaluate how far data from short- and long-term laboratory incubations may be used to predict *in situ*  $CO_2$  and methane fluxes, and (c) quantify the amount of permafrost OM released as  $CO_2$  during one year. Therefore, we measured *in situ*  $R_h$  fluxes of  $CO_2$  and methane in July 2016 and July 2019 from different soils in an active thaw slump of the northern Siberian Lena River Delta and related these to the thawed carbon pools in the active layer. In parallel, we incubated soil samples from the same sites and used the resulting  $CO_2$  and methane production rates to calculate potential GHG fluxes. Finally, we calibrated a regression model with *in situ* flux observations, and a dynamic model with long-term incubation data from samples of the same permafrost deposits, to estimate annual  $CO_2$  production from the thawed permafrost and to determine the fraction of permafrost OM decomposed over a whole year.





**Figure 1.** Overview of the thaw slump on the island Kurungnakh and soil profiles of four sampling sites. (a) Areal view of the thaw slump with non-vegetated TMs (light gray, middle) surrounded by relocated and mixed material containing also the surface vegetation cover of the former active layer (green patches). (b) Headwall of the thaw slump with massive ice wedges (bright gray), the mineral material of the former polygonal centers (dark brown) and eroding active layer material at the top and at the foot of the headwall. Soil profiles of the active layer at the sampled sites (c) SF3, (d) SF4, and (e) SF5 with visible inclusions of recent OM from surface vegetation (brownish patches), and (f) TM2 without distinct inclusions of organic material. Black scale bar in panels c to f corresponds to 10 cm.

# 2. Materials and Methods

#### 2.1. Study Sites

Field measurements and sampling were conducted at an active thaw slump on the island Kurungnakh (72.339°N, 126.292°E, Figure 1) in the Lena River Delta, North Siberia. The investigation area is situated in the zone of continuous permafrost with a trans-arctic, continental climate. The annual mean air temperatures on the neighboring island Samoylov was -12.3 °C with annual soil temperatures ranging between about  $10^{\circ}$ C and  $-28^{\circ}$ C at 6 cm depth (Boike et al., 2019). The soils are frozen for about eight months of the year. Only during the short summer season, a shallow surface layer thaws to a depth of less than 70 cm. Kurungnakh is comprised of Yedoma deposits (17-34 m above river level) that developed during the middle and late Weichselian (MIS 3 and MIS 2) and contain huge syngenetic ice-wedges. The OM frozen in these deposits originate mainly from a grass-dominated tundra-steppe vegetation. The Yedoma is overlain by an about 5 m thick cover of Holocene deposits. The mineral soil is mainly comprised of sandy to silty loam with intermediate organic carbon contents (2%-7%) (Knoblauch et al., 2013; Wetterich et al., 2008). Due to their high ice content, Yedoma deposits are strongly affected by thermokarst processes and particularly vulnerable to abrupt thaw (Czudek & Demek, 2017). After thawing of massive ice-wedges exposed at the bluffs of the Lena River channels, retrogressive thaw slumps develop. Such retrogressive thaw slumps are a widespread form of thermokarst and one of the most rapidly developing erosional features (Bröder et al., 2021; Cassidy et al., 2016; Tanski et al., 2017). In front of the thaw slump's headwall, soil material is relocated and forms a mud-slurry, which is then transported downslope. Due to rapid erosion, the surface of an active thaw slump is bare of recent vegetation and only  $R_{\rm h}$  contributes to in situ CO<sub>2</sub> and methane



#### Table 1

Position and Characteristics of the Studied Thaw Slump Floor Soils (SF) and the Thermokarst Mound Soils (TM) Sampled in 2016 and 2019

		2016						2019				
Site	Position	depth (cm)	pН	TOC (%)	C/N	<sup>a</sup> TOC Pool (kg m <sup>-2</sup> )	<i>mcrA</i> gene copies (x 10 <sup>5</sup> g <sup>-1</sup> )	depth (cm)	pН	TOC (%)	C/N	<sup>a</sup> TOC pool (kg m <sup>-2</sup> )
SF1	72.3390 N 126.2929 E	0-20	7.0	1.20	11	6.2	$2.91 \pm 1.94$	<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.
		20-42	7.0	2.23	13		$43.0 \pm 31.1$	<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.	<sup>a</sup> n.d.	
								<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.	<sup>a</sup> n.d.	
SF3	72.3390 N 126.2921 E	0-20	6.5	3.71	16	15.2	$31.4 \pm 23.8$	0–10	$5.5 \pm 0.1$	$6.1 \pm 2.5$	$19 \pm 0.8$	$15.2 \pm 1.5$
		20-42	6.7	4.82	16		$43.6 \pm 40.3$	10-20	$5.5 \pm 0.2$	$3.8 \pm 0.13$	$20 \pm 0.2$	
								20-30	$5.5 \pm 0.1$	$4.0\pm0.05$	$20 \pm 0.7$	
SF4	72.3388 N 126.2913 E	0-20	6.4	4.67	16	14.8	$19.7 \pm 2.89$	<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.
		20-43	6.6	3.80	16		$8.46 \pm 1.03$	<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.	<sup>a</sup> n.d.	
								<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.	<sup>a</sup> n.d.	
SF5	72.3391 N 126.2919 E	0-20	5.7	4.35	17	°15.0	$2.36 \pm 0.85$	<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.
		20-40	5.7	3.62	16		$3.54 \pm 1.20$	<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.	<sup>a</sup> n.d.	
								<sup>b</sup> n.d.	<sup>a</sup> n.d.	<sup>b</sup> n.d.	<sup>a</sup> n.d.	
SF6	72.3390 N 126.2914 E	0-17	6.5	4.35	16	10.6	<sup>b</sup> n.d.	0–10	$5.8 \pm 0.1$	$2.5\pm0.09$	$19 \pm 0.6$	$10.1\pm0.9$
								10-20	$5.4 \pm 0.0$	$3.1\pm0.90$	$20 \pm 0.8$	
								20-30	5.4(n = 1)	3.8(n = 1)	21 (n = 1)	
TM1	72.3393 N 126.2929 E	0-20	6.5	2.87	11	<sup>d</sup> 11.5	$1.92 \pm 0.43$	0–10	$7.9\pm0.1$	$3.2 \pm 0.11$	$14 \pm 0.1$	$11.6\pm0.5$
								10-20	$7.9\pm0.1$	$3.9 \pm 0.28$	$13 \pm 0.2$	
								20-30	$7.7\pm0.0$	$3.4 \pm 0.20$	$13 \pm 0.0$	
TM2	72.3393 N 126.2919 E	0-20	7.9	2.35	12	8.9	0.08 (n = 1)	0–10	$8.1\pm0.1$	$1.6\pm0.13$	$15 \pm 0.3$	$7.2 \pm 0.4$
		20-43	7.9	1.98	11		0.04 (n = 1)	10-20	$8.2\pm0.0$	$1.6\pm0.05$	$14 \pm 0.4$	
								20-30	$8.2 \pm 0.0$	$1.8 \pm 0.25$	$14 \pm 0.2$	

<sup>a</sup>calculated for a soil depth of 30 cm. <sup>b</sup>n.d. = not determined. <sup>c</sup>calculated with mean bulk density of SF sites. <sup>d</sup>calculated with mean bulk density of TM2, assuming constant TOC concentrations in the uppermost 30 cm.

fluxes, which makes these features ideal systems for studying the *in situ* decomposition of recently thawed permafrost OM.

We measured *in situ* gas fluxes in July 2016 at seven field sites in such an active thaw slump (Table 1 and Figure S1 in Supporting Information S1). The sites were located in an area of about 0.5 ha with a distance of about 30 m to each other and were grouped according to their parent soil material. Five sites were sampled at the thaw slump floor (SF) and consisted of relocated soil material, which mainly comprised of recently thawed permafrost but also some surface mineral soil material and remnants of surface vegetation from active layer detachment at the front of the headwall (Figures 1c–1e). Further two sites were sampled at two thermokarst mounds (TM) that were not affected by relocated surface material and consisted of mineral soil of the former polygonal centers between the ice wedges (Figures 1b and 1f). In July 2019, two SF sites (SF3 and SF6) and the two TM sites were resampled.

#### 2.2. Field Measurements and Sampling

 $CO_2$  and methane fluxes were measured about every other day between July 12, 2016 and July 28, 2016 and between July 10, 2019 and July 23, 2019 with opaque emission chambers (diameter 24 cm, 20 cm height). Three soil collars were pre-installed at each site at least 24 hr before the first measurement. At each sampling day, three emission measurements were conducted above these three different soil collars per site.  $CO_2$  concentrations inside the chambers were quantified over a period of 5 min by portable GHG analyzers



using a Li-840a (LI-COR Biosciences, Lincoln, USA) during the first week of measurements and an UGGA (Los Gatos Research, San Jose, USA) during the second week in 2016. Methane emissions were measured during the first week by collecting gas samples over a period of 30 min from the closed chambers that were analyzed the same day by gas chromatography (Preuss et al., 2013). During the second week, methane concentrations were quantified with the UGGA in parallel to  $CO_2$  concentration measurements. Both analyzers were calibrated at the field station on Samoylov with an external calibration gas (403 ppm  $CO_2$ , 3.1 ppm  $CH_4$ ). In parallel to each gas flux measurement, thaw depth, air temperature and soil temperature gradients (5-cm increments between surface and thaw depth) were measured. In July 2019,  $CO_2$  and methane fluxes were measured with an UGGA, air and soil temperatures and thaw depth as described above.

Soil samples were collected in 2016 and 2019 between the soil surface and the thaw depth using steel tubes (inner diameter 5.8 cm). In 2016, one profile was sampled at each sampling site at the beginning of the measurement period (July 10, 2016). Two soil samples were retrieved per site, the first between the surface and 20 cm depth and the second between 20 cm and the thaw depth (Table 1). In 2019, three different profiles were sampled per sampling site with a 10 cm depth increment between the surface and the thaw depth in the middle of the measurement period (July 18, 2019). All soil samples were quantitatively retrieved from the sample tube and the bulk density was calculated from sample volume and dry weight.

#### 2.3. Soil Analysis

Total soil carbon and nitrogen were quantified with an elemental analyzer (VarioMAX Elementar Analysensysteme GmbH, Hanau, Germany) after the soil had been sieved (<2 mm), milled, and dried at 105°C. Total inorganic carbon (TIC) was quantified from the amount of  $CO_2$  released after sample treatment with phosphoric acid. Total organic carbon (TOC) was calculated as the difference between total carbon and TIC. Soil pH was measured in a suspension of 10 g of dried soil in 25 mL of demineralized water. Soil water content was calculated from the weight difference between fresh samples and those dried at 105°C. The <sup>14</sup>C-age of bulk TOC was analyzed with a MICADAS AMS system (Ionplus, Dietikon, Switzerland) (Steinhof et al., 2017). AMS <sup>14</sup> C data were calibrated using Oxcal 4.3 and the INtCal13 calibration curve (Bronk Ramsey, 2009).

#### 2.4. Incubation Experiments

Short-term laboratory incubations were conducted with samples from sites SF1, SF3, SF4, SF6 and TM2 (Table 1) in July 2016. Samples were processed at the day of sample collection and incubated at the research station on Samoylov Island. For anaerobic incubations, samples were weighted into glass flasks (100 mL) in a glove bag under nitrogen. The bottles were closed with rubber stoppers and the headspace was repeatedly evacuated and flushed with nitrogen. Aerobic incubations were prepared under ambient air in 250-mL bottles. All samples were incubated in three replicates for 18 days at 4°C. The amount of  $CO_2$  and methane was measured every day (aerobic incubations) or every other day (anaerobic incubations) with a gas chromatograph (Knoblauch et al., 2018).  $CO_2$  and methane production rates were calculated from cumulative gas production during the last two weeks of the incubations, to omit initial disturbances during sample preparation. Samples from the surface soil of TM2 were incubated at 4°C for two years and reanalyzed in July 2018.

#### 2.5. Calculation of Carbon Dioxide and Methane Fluxes

Methane and CO<sub>2</sub> emissions from bare soils were calculated by a linear regression of gas concentrations inside the emission chambers as described by Eckhardt and Kutzbach (2016). Gas emissions were normalized to surface area ( $f_A$ ).

Field  $CO_2$  fluxes were also normalized to the amount of organic carbon thawed in the active layer at the day of gas flux measurements, using the carbon concentrations in the different sampled soil layers according to:

$$C_{A} = \sum_{i=1}^{n} \rho_{i} \cdot d_{i} \cdot \left[ \text{TOC} \right]_{i}$$
(1)

with  $C_A = Carbon pool in the active layer, \rho_i = dry bulk density of soil layer$ *i* $, d_i = depth of soil layer$ *i* $and <math>[TOC]_i = TOC$  concentration in layer *i*.

The  $CO_2$  or methane flux normalized to thawed carbon during each of the emission measurement ( $f_c$ ) was calculated according to:

$$f_C = \frac{f_A}{C_A}.$$
 (2)

To evaluate how far results from short-term laboratory incubations represent GHG production from thawing permafrost OM under *in situ* conditions,  $CO_2$  and methane production rates from laboratory incubations were upscaled to a potential areal flux ( $f_p$ ), which was calculated for each day of field flux measurements according to:

$$f_{P} = \sum_{i=1}^{2} \left( \rho_{i} \cdot d_{i} \cdot r_{i} \cdot Q_{10}^{\frac{(T_{i}-4)}{10}} \right)$$
(3)

with  $r_i = aerobic$  or anaerobic CO<sub>2</sub> or methane production rate determined at 4°C in samples incubated from layer *i*,  $T_i =$  mean temperature of layer *i* during the day of field CO<sub>2</sub> and methane flux measurement.  $Q_{10}$  is the factor by which a reaction rates increases after a temperature rise of 10°C. A  $Q_{10}$  value of 2 was applied, which is based on permafrost incubation studies (Schädel et al., 2016; Vaughn & Torn, 2019).

#### 2.6. Soil Temperature Modeling

To extrapolate summer  $CO_2$  emissions to an annual  $CO_2$  flux, a year-round time series of soil profile temperature data is required. This time series was derived for 5 cm layers until 1 m soil depth using the land surface model JSBACH (Beer et al., 2018; Porada et al., 2016) with a specific setup for the Lena Delta (Chadburn et al., 2017; Ekici et al., 2015) but without considering any vegetation insulation layer (e.g., mosses) as the sampling sites were not vegetated. Only for model evaluation, we additionally compared a model run with total moss cover to the soil temperature data from Samoylov Island nearby. The meteorological forcing data is based on the CRUNCEP dataset (Viovy, 2018). This time series was bias-corrected to observations from the Samoylov station (Boike et al., 2019) using an overlap period 2003–2016 and an additive method except for precipitation and wind speed, for which a multiplicative technique was applied (Beer et al., 2014). Then, climate data were replaced by observed meteorological data from the Samoylov station (Boike et al., 2019) during 2002–2016. Winter precipitation was approximated by positive day-to-day snow depth changes. Soil water, ice content and temperature were initialized using a classical spin-up procedure following Beer et al. (2018), followed by a transient model run from 1901 to 2016. Soil temperature results with and without moss layer were evaluated with observations at the Samoylov station (with moss layer) and by direct thaw slump measurements during July 2016 (without moss layer).

#### 2.7. Annual Carbon Dioxide Fluxes of the Thaw Slump

The annual  $CO_2$  fluxes from heterotrophic respiration averaged over the different thaw slump sites were estimated using two different modeling strategies and by using different observations. First, a  $Q_{10}$  model was calibrated against July 2016 chamber measurements of  $CO_2$  emissions and observed average topsoil (0–20 cm) temperature. Subsequently, the  $Q_{10}$  model was applied to year-round topsoil temperature estimates from a site-level run of the land surface scheme JSBACH (Section 2.6). Second, a first-order kinetics model (ICBM) with two connected reservoirs (Andrén & Kätterer, 1997) was applied. The parameters at a reference temperature of 4°C were calibrated using long-term incubation data from either Holocene or Pleistocene deposits nearby (Knoblauch et al., 2013). The decomposition rate constants are modified by a  $Q_{10}$  model using a  $Q_{10}$  value of 2 (Schädel et al., 2016; Vaughn & Torn, 2019). The ICBM was driven by soil temperature data (20 five-cm-layers) from the site-level JSBACH model run (Section 2.6) but with observed soil OM content (Table 1) for topsoil and subsoil layers from each site. This approach led to several results per thaw slump site due to a variation of calibrated decomposition rate constants and results in a distribution of CO<sub>2</sub> fluxes from which we present the median and the upper and lower quartiles. Model outputs of



the ICBM were considered at temperatures down to  $-10^{\circ}$ C since below this temperature, microbial activity was shown to be practically negligible (e.g., Mikan et al., 2002; Natali et al., 2019).

#### 2.8. Quantification of Methanogenic Archaea

Total genomic DNA was extracted in duplicates using the DNeasy Power Soil Kit (Qiagen, Hilden, Germany) according to the manufacturer's protocol. Quality and quantity of the DNA was assessed through gel electrophoresis and photometry using the Qubit2 system (Invitrogen, HS-quant DNA). DNA was purified using the 15HiYield PCR Clean-Up & Gel-Extraction Kit (SLG, Gauting, Germany) to reduce PCR inhibitors prior to PCR applications. The enumeration of methanogenic gene copies was realized through quantitative PCR (qPCR) targeting the *mcrA* gene of methanogens as described by Liebner et al. (2015). Each of the duplicate DNA extractions per sample was amplified in triplicates.

#### 2.9. Statistics

Data were tested for normal distribution by the Shapiro-Wilk test. Since data transformation (log10 and square root) of non-normally distributed data did not improve data distribution, statistical tests were performed with non-transformed data. In case of normal distribution, differences between mean values were compared by a two-sided independent *t*-test and correlations between variables by Pearson's correlation coefficient. In case of nonnormal distribution, non-parametric tests were performed, for two groups a Mann-Whitney U test and for more than two groups an independent samples Kruskal-Wallis test followed by a Dunn-Bonferoni post-hoc test. A Spearman's rho correlation analysis with a two tailed significance test was applied to test correlation between non-normal distributed data. Significance was tested on a level of p < 0.05. All statistical analyses were conducted with IBM SPSS Statistics 27 (IBM Corporation, Armonk, USA).

### 3. Results

#### 3.1. Soil Properties

All soils were classified as Protic Cryosols due to the presence of permafrost and a lack in soil horizon development (WRB, 2014) (Figure 1). Neutral pH values dominated with extreme values between moderately acidic at SF5 to moderately alkaline at TM2 (Table 1). Organic carbon concentrations ranged between 1.2% and 6.1% and C/N ratios between 11 and 21 and both variables showed no clear change with depth. The average TOC concentration of SF soils ( $3.6 \pm 1.2\%$  (mean  $\pm$  SD) in 2016,  $3.8 \pm 1.2\%$  in 2019) was substantially higher than in TM soils ( $2.4 \pm 0.4\%$  in 2016,  $2.6 \pm 1.0\%$  in 2019) but this difference was not significant (*t*-test, p > 0.07). Except for one site (TM2), TOC pools in the uppermost 30 cm did not differ between 2016 and 2019 (Table 1). However, due to substantially higher air temperatures during the thaw period, the soil temperature was higher in 2019 than in 2016 (Table S1 in Supporting Information S2) with significant differences at SF6 and TM1 (*t*-test, p < 0.01). Consequently, both thaw depth and thawed carbon pools were significantly higher (Mann-Whitney U test, p < 0.005) at all sampling sites in 2019 in comparison to 2016. The organic carbon originated from the Pleistocene/Holocene transition (10-13 kyr) at SF1, the site most distant from the headwall, while TOC of the remaining SF sites were of mid to late Holocene age (Table S2 in Supporting Information S2). In contrast, OM at the TM sites originated from the last glacial maximum (TM2) and the Kargin interstadial (TM1) with a maximum age of 33 kyr BP. Only SF soils contained mixedin organic rich material of modern <sup>14</sup>C-age (see SF4 Table S2 in Supporting Information S2). Soil moisture did not clearly differ in July 2019 and July 2016 (Table S3 in Supporting Information S2).

#### 3.2. Abundance of Methanogenic Archaea

The abundance of methanogenic archaea was significantly higher (Mann-Whitney U test, p < 0.01) in the SF soils than in TMs (Table 1). Highest abundance of methanogens was found in the bottom samples of SF3 and lowest in the bottom layer of TM2. The number of *mcrA* genes was below the detection limit of about thousand copies per gram in 61% of the TM soil amplifications while the qPCR was successful in all of the replicates of the SF soil samples. *mcrA* gene copy numbers correlated significantly with CH<sub>4</sub> production





**Figure 2.** (a) *In situ*  $CO_2$  fluxes ( $f_A$ ) in July 2016 and July 2019 at a thaw slump on the island Kurungnakh and (b) calculated potential fluxes ( $f_p$ ) using aerobic and anaerobic  $CO_2$  production rates from short-term incubations in 2016. The boxes give the upper and lower quartiles, the whiskers the 95 and 5 percentiles, the horizontal lines stand for the median values and the closed circle show the arithmetic means. In 2019, gas fluxes were measured only at sites SF3, SF6, TM1, and TM2.

(Spearman's rho = 0.79, p = 0.007) and anaerobic CO<sub>2</sub> production (Spearman's rho = 0.69, p = 0.029) in laboratory incubations but with none of the soil parameters tested (TOC, N, C:N, pH, water content). Furthermore, methanogen abundance in the active layer did not correlate with methane emissions from the respective sampling sites (Spearman's rho = -0.185, p = 0.265). The undisturbed active layer soil above the headwall contained in its bottom mineral horizon (8–20 cm depth) 26.0 ± 20.9 × 10<sup>5</sup> mcrA gene copies per gram.

#### 3.3. Carbon Dioxide and Methane Fluxes

Median CO<sub>2</sub> fluxes from the different thaw slump soils in 2016 and 2019 ranged between 0.24 and 2.6 g CO<sub>2</sub>-C m<sup>-2</sup> d<sup>-1</sup> (Figure 2a). Only at SF3, median fluxes were significantly lower in 2019 in comparison to 2016 (Mann-Whitney U test, p < 0.001) while at the other sites, CO<sub>2</sub> fluxes in 2019 were either slightly higher (SF6, TM1) or similar (TM2) than in 2016 with no significant differences between the two sampling years (Figure 2a). Median CO<sub>2</sub> emissions from all SF sites (0.89 g CO<sub>2</sub>-C m<sup>-2</sup> d<sup>-1</sup> in 2016 and 2019) were not significantly different (Mann-Whitney U test, p > 0.18) to those from the TM sites (0.84 and 0.74 g CO<sub>2</sub>-C m<sup>-2</sup> d<sup>-1</sup> in 2016 and 2019, respectively) (Table S4 in Supporting Information S2). The overall median of CO<sub>2</sub>

flux from all sites was 0.89 g CO\_2-C m^{-2} d^{-1} in 2016 and 0.83 g CO\_2-C m^{-2} d^{-1} in 2019.



**Figure 3.** Carbon dioxide emissions normalized to the amount of thawed carbon  $(f_C)$  at the different sampling sites. Boxes give the upper and lower quartile, the whiskers the 95 and 5 percentile, the horizontal line the median and the closed circle the mean value.

In 2016, CO<sub>2</sub> fluxes per square meter were significantly correlated (Spearman's rho 0.32 to 0.38, p < 0.01) with the average active layer temperature, TOC (Figure S2a in Supporting Information S1) and total nitrogen in the active layer, while in 2019, CO<sub>2</sub> fluxes significantly correlated only with the amount of TOC and nitrogen in the active layer during the day of measurement.

When normalizing CO<sub>2</sub> fluxes to the amount of unfrozen TOC in the active layer during flux measurements, median fluxes at the different sites ranged between 0.020 and 0.118 g CO<sub>2</sub>-C (kgC)<sup>-1</sup> d<sup>-1</sup> and 0.015 and 0.067 g CO<sub>2</sub>-C (kgC)<sup>-1</sup> d<sup>-1</sup> in 2016 and 2019, respectively, with highest CO<sub>2</sub> fluxes from TM1 in both years (Figure 3). Despite substantially higher soil temperatures in 2019 (Table S1 in Supporting Information S2), median fluxes from all sites decreased (Mann-Whitney U test, p < 0.001) from 0.061 g CO<sub>2</sub>-C (kgC)<sup>-1</sup> d<sup>-1</sup> in 2016 to 0.033 g CO<sub>2</sub>-C (kgC)<sup>-1</sup> d<sup>-1</sup> in 2019 (Table S4 in Supporting Information S2). Also the site wise comparison showed







**Figure 4.** In situ methane fluxes ( $f_A$ ) measured in the field in July 2016 and July 2019 at a thaw slump on Kurungnakh (a) and potential fluxes ( $f_p$ ) calculated using anaerobic methane production rates from short-term incubations (b). The boxes give the upper and lower quartile, the whiskers the 95 and 5 percentile, the horizontal lines stand for the median values and the closed circle show the arithmetic means. In 2019, gas fluxes were measured only at the sites SF3, SF6, TM1, and TM2.

a decrease of CO<sub>2</sub> fluxes at each site with a significant decrease at SF3, TM1 and TM2 (Mann-Whitney U test, p < 0.005). Furthermore, median CO<sub>2</sub> fluxes normalized to thawed carbon did significantly differ (Mann-Whitney U test, p < 0.001) between SF sites (0.053 g CO<sub>2</sub>-C (kgC)<sup>-1</sup> d<sup>-1</sup>) and TM sites (0.094 g CO<sub>2</sub>-C (kgC)<sup>-1</sup> d<sup>-1</sup>) in 2016 but not in 2019.

In 2016, median methane fluxes ranged between 3.7 and 5.1 mg  $CH_4$ -C m<sup>-2</sup> d<sup>-1</sup> at the four sites that showed significant methane emissions (SF3, SF4, SF5, and SF6) (Figure 4a). Methane fluxes from the three replicate plots of the same site could differ substantially, e.g., between 132 and 0.8 mg  $CH_4$ -C m<sup>-2</sup> d<sup>-1</sup> at the same day at SF4. Methane fluxes at SF3 correlated significantly with thaw depth in 2016, and thawed carbon and nitrogen during measurement (Spearman's rho = -0.717, p < 0.001) but no significant correlation was found at any other site with any environmental parameter tested (thaw depth, surface and mean soil temperature, amount of thawed TOC and nitrogen). In 2019, methane fluxes from SF6, the only site releasing considerable amounts of methane, were positively correlated with thaw depth, and thawed carbon and nitrogen (Spearman's rho = 0.488, p = 0.001). Median methane fluxes at SF6 were higher in 2016 (4.8 mg  $CH_4$ -C m<sup>-2</sup> d<sup>-1</sup>) than in 2019 (2.8 mg  $CH_4$ -C m<sup>-2</sup> d<sup>-1</sup>) and this difference was close to significant (Mann-Whitney U test, p = 0.067) (Figure 4a). At sites with detectable methane emissions, methane contributed between 0.02% and 6.5% (median 0.41% and 0.11% in 2016 and 2019, respectively) to total carbon fluxes to the atmosphere.

#### 3.4. Carbon Dioxide and Methane Production in Laboratory Incubations

Microbial CO<sub>2</sub> production rates under aerobic conditions were highest in surface samples of SF3 and lowest in the surface sample of SF1 (Figure S3 in Supporting Information S1). The median value of aerobic CO<sub>2</sub> production rates from all sampling sites was 0.73 (0.82, 0.41 upper and lower quartile)  $\mu$ mol CO<sub>2</sub> g<sup>-1</sup> d<sup>-1</sup>. Anaerobic conditions slowed down CO<sub>2</sub> production in all except for one soil layer (TM2, 20–43 cm) by an average factor of 3.8 ± 2.6 (*n* = 9) (Figure S3 in Supporting Information S1). Aerobic CO<sub>2</sub> production was most strongly correlated with TOC and total nitrogen (Spearman's rho = 0.69, *p* < 0.001) and the C:N ratio (Spearman's rho = 0.68, *p* < 0.001). Anaerobic CO<sub>2</sub> production rates did not correlate with any of these parameters.

Anaerobic methane production was observed in all samples except for those from TM2, but the rates were about three orders of magnitude lower than anaerobic  $CO_2$  production rates (median  $CO_2$ :CH<sub>4</sub> ratio 657 (215, 4,707) (upper and lower quartile)) with highest methane production in the surface samples of SF1 and SF3 and lowest in SF6 (Figure S2c in Supporting Information S1). However, after two years of incubation





**Figure 5.** Evaluation of JSBACH topsoil temperature in 2016 at the thaw slump without vegetation cover and at a polygon on the nearby island Samoylov (average of polygon rim, slope and center) with vegetation cover. Topsoil temperature represents the average temperature in the uppermost 20 cm of soil. The observational data from Samoylov are from Boike et al. (2019).

at 4°C on Samoylov, methane production could also be detected in the TM2 surface soil with a final mean ratio between total  $CO_2$  and methane production of 5.9 and 6.9 (n = 2).

#### 3.5. Potential Gas Fluxes From Laboratory Incubations

Potential CO<sub>2</sub> fluxes ( $f_p$ ), calculated from CO<sub>2</sub> production rates in aerobic short-term incubations significantly (Kruskall-Wallis, p < 0.01) overestimated *in situ* CO<sub>2</sub> fluxes by a factor of 1.3–6.4 (mean 4.4 ± 1.4 SD) (Figure 2). Since anaerobic conditions substantially reduced CO<sub>2</sub> production rates, the difference between *in situ* and potential CO<sub>2</sub> fluxes decreased with significantly higher potential fluxes at sites SF1 and TM2 (Kruskall-Wallis, p < 0.02), similar fluxes at SF4 and significantly lower potential fluxes at SF3 and SF6 (p < 0.01). The mean ratio between *in situ* fluxes and potential anaerobic fluxes was  $1.9 \pm 1.5$ . *In situ* CO<sub>2</sub> fluxes from all sites correlated both with potential aerobic CO<sub>2</sub> fluxes (Spearman's rho = 0.62, p < 0.001, Figure S2b in Supporting Information S1) and with potential anaerobic CO<sub>2</sub> fluxes (Spearman's rho = 0.28, p < 0.05).

Potential methane fluxes from laboratory incubations were significantly lower (Mann-Whitney-U test, p < 0.002) than in situ emissions at sites SF4 and SF6 (Figure 4b). Since no significant methane emissions could be detected at SF1, potential methane fluxes overestimated *in situ* emissions at this site while potential and *in situ* fluxes did not significantly

differ at site SF4. At TM2, we detected no methane production in short-term laboratory incubations and also no methane emissions from the soils.

#### 3.6. $Q_{10}$ Model Calibration

The  $Q_{10}$  regression model could only be calibrated at sites SF1, SF3, and SF4 (Figure S4 in Supporting Information S1) since the *in situ* soil temperature range during flux measurements was too small at the other sites.  $Q_{10}$  values were very similar at SF1 and SF3 (2.7 and 2.6, respectively) but substantially higher



**Figure 6.** Comparison of daily emission results between both models for 2016 and chamber measurements in July 2016. Chamber measurements from sites used for  $Q_{10}$  model calibration (SF1, SF3, SF4) are shown while model results represent the median of estimates from different sites ( $Q_{10}$  model and ICBM) and parameter values (ICBM).

*a* soil temperature range during flux measurements was too small at the similar at SF1 and SF3 (2.7 and 2.6, respectively) but substantially higher at SF4 (4.1). In contrast,  $CO_2$  fluxes at 8°C were most similar at SF1 and SF4 (0.92 and 0.94 g  $CO_2$ -C m<sup>-2</sup> d<sup>-1</sup>) and more than twice as high (2.58 g  $CO_2$ -C m<sup>-2</sup> d<sup>-1</sup>) at SF3, although sites SF3 and SF4 have similar OM content, which were higher than at site SF1.

#### 3.7. Model Evaluation

The estimated year-round soil temperature dynamics simulated by the site-level run of the JSBACH land surface scheme without a moss layer shows a high agreement with observations at the thaw slump in July 2016 (Figure 5). Additionally, the JSBACH model results considering a 90% moss cover show a very good agreement with the observations from a vegetated polygon at nearby island Samoylov (Figure 5). There is a positive temperature bias in December 2016 when comparing the model simulation including a moss layer to the temperature data of the Samoylov soil station. The resulting positive bias of estimated CO<sub>2</sub> emission during December 2016 is about 3% ( $Q_{10}$  model) and 5% (ICBM).

The CO<sub>2</sub> emissions from the thaw slump in 2016 simulated with the  $Q_{10}$  model and the ICBM show a high agreement with *in situ* chamber measurements (Figure 6). The ICBM results, based on both topsoil and subsoil temperature, show less day-to-day variation than the  $Q_{10}$  model that purely relates emissions to temperature and therefore translate all





**Figure 7.** Cumulative  $CO_2$  emission from the thaw slump in 2016 calculated with the  $Q_{10}$  model and the ICBM (a) and fraction of thawed total organic carbon (TOC) mineralized simulated by the ICBM (b). The solid line give the median values and the shaded area the upper and lower quartiles of the  $Q_{10}$  model (red) and the ICBM (blue).

temperature variation immediately into emission variation. However, both model results are similar on average after July 15 (bias = -0.03 g CO<sub>2</sub>-C m<sup>-2</sup> d<sup>-1</sup> and RMSE = 0.15 g CO<sub>2</sub>-C m<sup>-2</sup> d<sup>-1</sup>), although both models are calibrated using completely independent datasets (chamber measurements vs. long-term incubation experiments) and are driven by independent data (topsoil temperature vs. soil OM content and total soil temperature column dynamics).

#### 3.8. Annual Carbon Dioxide Emissions From the Thaw Slump

The median cumulative annual carbon release as  $CO_2$  from the thaw slump in 2016 was estimated by the two models as 184 (269,116) g $CO_2$ -C m<sup>-2</sup> (ICBM) and 160 (342, 139) g $CO_2$ -C m<sup>-2</sup> ( $Q_{10}$  model) (Figure 7a). These are the median and (75, 25)-percentiles from a range of model results using information from the different thaw slump sites ( $Q_{10}$  model) or a calibrated parameter distribution (ICBM). Since simulated surface soil temperatures were below  $-10^{\circ}$ C between January 1, 2016 and May 24, 2016 (Figure 5), zero  $CO_2$  production was assumed in this time period by the ICBM. Soil temperatures below  $0^{\circ}$ C were simulated for the topsoil layer by the JSBACH land surface scheme from of January 1 until May 28 and after September 24, 2016 and the  $CO_2$  fluxes during this period accounted for 25% (ICBM) to 31% ( $Q_{10}$  model) of annual  $CO_2$  fluxes.

When relating these emissions to the available soil OM thawed in the active layer at the respective day of the year, the ICBM estimates that 0.78 (1.16, 0.52)% of thawed soil OM was mineralized to  $CO_2$  over the whole year 2016 (Figure 7b).

# 4. Discussion

#### 4.1. In Situ Carbon Dioxide and Methane Fluxes

Abrupt permafrost thaw is affecting only a minor fraction of the northern permafrost region but GHG fluxes from these areas are expected to almost equal those from the remaining permafrost region until 2300 (Turetsky et al., 2020). Since the studied thaw slump soils were not vegetated, GHG emissions originated only from recently thawed permafrost and do not consider plant autotrophic CO<sub>2</sub> respiration like most previous studies (Abbott & Jones, 2015; Natali et al., 2015; Ravn et al., 2020). The heterotrophic CO<sub>2</sub> fluxes at Kurungnakh (0.24–2.6 g CO<sub>2</sub>-C m<sup>-2</sup> d<sup>-1</sup>) were relatively high in comparison to the median CO<sub>2</sub> emissions from soils of the same region, e.g., from similar thawing Yedoma deposits (0.07–0.43 g CO<sub>2</sub>-C m<sup>-2</sup> d<sup>-1</sup>) (Vonk et al., 2012), or different polygonal tundra soils (0.24–0.48 g CO<sub>2</sub>-C m<sup>-2</sup> d<sup>-1</sup>) (Eckhardt et al., 2019). However, since soil temperature and active layer carbon pools were the main driving parameters for CO<sub>2</sub> exchange fluxes, comparing gas fluxes from different tundra soils without knowledge of latter parameters is problematic. Similar heterotrophic CO<sub>2</sub> fluxes as on Kurungnakh were only measured from active thaw



slump soils in north-western Alaska (Jensen et al., 2014) but at substantially higher soil temperatures and active layer depths. The relatively high heterotrophic respiration fluxes from the Kurungnakh thaw slump soils indicate the presence of highly labile OM but laboratory and field incubations indicate that this labile carbon pool comprises only a minor fraction of TOC (Knoblauch et al., 2013; Moni et al., 2015; Schädel et al., 2014; Zimov et al., 2006). Therefore, it was expected, that  $CO_2$  fluxes would decline over time. This was also the case after normalizing *in situ*  $CO_2$  fluxes to the amount of organic carbon in the active layer. Despite substantially higher soil temperatures in 2019, significantly lower  $CO_2$  fluxes per thawed carbon were observed than in 2016, which indicates a decreasing degradability of the remaining organic carbon in the active layer. Also surface erosion, causing a loss of labile carbon, might have contributed to lower  $CO_2$  fluxes in 2019 than in 2016.

Mean CO<sub>2</sub> fluxes normalized to thawed organic carbon were lower at the SF soils than at the TM soils, both under *in situ* conditions and in laboratory incubations. This is surprising, since the SF soils contained fresh OM from the surface vegetation of the eroding headwall, and it has been shown that fresh OM may increase OM decomposition rates in thawing permafrost (positive priming) (Walz et al., 2017; Wild et al., 2016). The reason for lower decomposition rates in the SF soils might be a higher disturbance due to strong erosion, which was less pronounced at the TM sites. The higher decomposability of the OM at the TM sites is further striking considering its substantially higher age (20–33 kyr BP) than the OM at the SF sites (2–13 kyr BP). These results support data from laboratory incubations that show no effect of organic carbon age on decomposition rates and a particularly high decomposability of organic carbon in the Siberian deposits of the MIS3 Kargin interstadial (Knoblauch et al., 2013; Walz et al., 2018). Thus, we present evidence that the permafrost OM in the Pleistocene sediments was well preserved during its storage for thousands of years under frozen conditions, and that microbial activity in the permafrost is likely of minor importance at least at the low average temperature of  $-10^{\circ}$ C in the deep permaforst (Boike et al., 2019). The latter conclusion is supported by multi-proxy studies indicating a low decomposition state of the Pleistocene Yedoma deposits in the study region (Strauss et al., 2015).

Methane emissions from the sampling sites were about two orders of magnitude lower than  $CO_2$  emissions and contributed only a minor amount to total carbon fluxes. Since methane is produced only in the absence of oxygen, its formation in soils is generally restricted to water saturated conditions. Furthermore, methane may be oxidized to  $CO_2$  by aerobic methane oxidizers when migrating through aerobic soils (Knoblauch et al., 2008; Zheng et al., 2018). Therefore, methane emissions depend on a variety of environmental parameters with lowest or even negative fluxes in well drained soils and highest fluxes from water saturated, organic soils vegetated by vascular plants (Christensen et al., 1995; Kwon et al., 2017; Natali et al., 2015; Olefeldt et al., 2013). As expected, mean methane fluxes from the unsaturated, bare soils of the thaw slump were in the lower range of those reported for wet arctic tundra soils in Siberia, Greenland and Alaska (20–250 mg CH<sub>4</sub>-C m<sup>-2</sup> d<sup>-1</sup>) (Kutzbach et al., 2004; Kwon et al., 2017; Ström et al., 2015; Vaughn et al., 2016) but, at least in 2016, higher than those from other drained arctic tundra soils (Davidson et al., 2016; Jørgensen et al., 2015; Kutzbach et al., 2004; Kwon et al., 2017). Furthermore, the large variability in methane fluxes with maximum values of up to 132 mg CH<sub>4</sub>-C m<sup>-2</sup> d<sup>-1</sup> demonstrate substantial heterogeneity of soil conditions but also a high degradability of the thawing permafrost OM even under anaerobic conditions.

The absence of methane emissions at SF1 and SF3 (2019) is striking since these sites had the highest *mrcA* gene copy numbers and produced methane in laboratory incubations. Hence, the most likely explanation for lacking methane fluxes from these sites is the complete oxidation of produced methane in the unsaturated surface soil (Knoblauch et al., 2008; Zheng et al., 2018). In contrast, the abundance of methanogenesis most detected in TM2 was lowest among all soils and in part even below the detection limit and no methanogenesis was detected in TM2 soil incubations. It has been shown that permafrost deposits, including those from the studied Kurungnakh sites, may contain low numbers of methanogens (Holm et al., 2020; Knoblauch et al., 2018), which may result in the observed very low or absent methane production in recently thawed permafrost (Chowdhury et al., 2015; Knoblauch et al., 2013; Waldrop et al., 2010; Walz et al., 2018). Not only methanogenesis but also other microbial functions, like nitrification, may got lost in Yedoma deposits over the millennia under frozen conditions, which constrain carbon and nitrogen cycling in these soils after thaw (Monteux et al., 2020). In the current thaw slump, the dispersal of methanogens from the active layer



above the headwall, which contained high numbers of methanogens, into the soils of the thaw slump seems essential for active methane production and methane emission, at least in the initial phase after thaw.

#### 4.2. Potential Carbon Dioxide and Methane Fluxes From Laboratory Incubations

The decomposability of permafrost OM is preferentially quantified with incubation studies, which are mostly done under aerobic conditions (Faucherre et al., 2018; Gentsch et al., 2018; Knoblauch et al., 2013; Schädel et al., 2016; Zimov et al., 2006). The obtained data are subsequently used in models of different scale and complexity for estimating organic carbon decomposition after permafrost thaw (Elberling et al., 2013; Knoblauch et al., 2015; MacDougall & Knutti, 2016; Schneider von Deimling et al., 2015; Wang et al., 2017).

The aerobic CO<sub>2</sub> production rates in laboratory incubations of Kurungnakh soils are in the middle of the range of rates reported for mineral permafrost material of the Eurasian Arctic, quantified by short-term incubation studies ( $\leq 6$  months) at temperatures below 10°C (Faucherre et al., 2018; Gentsch et al., 2018; Zimov et al., 2006). Anaerobic CO<sub>2</sub> production rates are rarely reported from low temperature (<10°C) incubations of thawed permafrost and ranged in Kurungnakh and Samoylov deposits between 0.02 and 0.20 µmol CO<sub>2</sub> g<sup>-1</sup> d<sup>-1</sup> (Knoblauch et al., 2013; Walz et al., 2017), which covers the range of anaerobic CO<sub>2</sub> production rates in the SF soils studied here.

It is still unclear to which extent laboratory incubation data represent *in situ* carbon decomposition rates in thawing permafrost. The current aerobic short-term incubations substantially overestimated in situ CO, fluxes. However, even well drained upland soils may contain anaerobic microsites enabling anaerobic microbial processes such as fermentation, denitrification and methane production (Sexstone et al., 1988; von Fischer & Hedin, 2002). The distribution of anaerobic microsites is strongly regulated by water filled pore space and microbial respiration in the soil (Keiluweit et al., 2016; Wagner, 2017). Field methane fluxes indicated anaerobic conditions at least in some parts of the soils. Since a fraction of CO<sub>2</sub> emissions originate from anaerobic OM decomposition, which is about three times slower than aerobic respiration (Knoblauch et al., 2018; Schädel et al., 2016), aerobic CO, production rates will overestimate in situ decomposition rates but may be used as the upper benchmark for potential permafrost OM decomposition. On the other hand, anaerobic decomposition rates from laboratory incubations should underestimate in situ CO, fluxes. However, this was only the case for two sites. The reason is likely the increased availability of labile OM after mixing the soil during the preparation of the incubation experiments. This known bias may be avoided by longer incubation studies whose data may then be used for constraining dynamic carbon decomposition models (Elberling et al., 2013; Knoblauch et al., 2013; Schädel et al., 2014). Such models may provide carbon decomposition rates similar to those under in situ conditions, which was demonstrated by using a two pool decomposition model (ICBM), calibrated with data from a three years incubation study with permafrost samples from the study site (Figure 6).

Surprisingly, *in situ* methane emissions were generally higher than potential methane emissions. The opposite was expected, since a fraction of the produced methane will be oxidized before emitted under *in situ* conditions but not in anaerobic incubations. Initially low methane production rates in incubations indicate that methane production is not accelerated by labile OM, as was observed for  $CO_2$  production (Knoblauch et al., 2013; Lee et al., 2012; Waldrop et al., 2010), but that the conditions for methanogenesis were not optimal at the beginning of the experiments (Chowdhury et al., 2015; Høj et al., 2007; Knoblauch et al., 2013; van Hulzen et al., 1999). Methanogenesis is a syntrophic process requiring a close cooperation with fermenting organisms under specific micro-environmental conditions (Schink, 1997). Disturbances during sample preparation, e.g., by mixing of the soil samples, may disrupt such consortia, change micro-environmental conditions and hence reduce methane production until stable conditions are re-established. Therefore, methane fluxes may generally be underestimated in short-term incubations.

#### 4.3. Total Annual Carbon Release

Soil temperature and the amount of thawed organic carbon were the main driving factors of heterotrophic respiration at the study sites. Therefore, the two independent models to extrapolate summer field observations to annual CO<sub>2</sub> fluxes were based on these two factors. The range of soil temperatures was large enough



to be used for the  $Q_{10}$  model calibration only at three sites and we could only use positive temperatures. However, our  $Q_{10}$  model gives similar results below 0°C as an exponential model fitted to a wide range of winter CO<sub>2</sub> fluxes (Natali et al., 2019). These results give confidence that our  $Q_{10}$  model can be used for simulating CO<sub>2</sub> fluxes at soil temperatures below those used for calibration. Furthermore, the obtained  $Q_{10}$  values (2.6–4.1) are in the typical range of  $Q_{10}$  values reported for the temperature response of CO<sub>2</sub> fluxes from permafrost ecosystems (Fouché et al., 2017; Natali et al., 2019; Wang et al., 2020). The simulation of active layer soil temperatures in 2016 with the JSBACH land surface model was very close to the observations at the nearby tundra on Samoylov when including an insulating moss cover into the model (Figure 5). Since no vegetation was present at our thaw slump sites, we used the simulated temperature data without a vegetation cover, which were very close to the observations during the field flux measurements. There was only a positive bias of JSBACH concerning the length of the zero-curtain period in autumn, which might result in higher CO<sub>2</sub> emissions simulated with the model than in the field.

Data on annual CO<sub>2</sub> emission fluxes from  $R_h$  of tundra soils are currently not available and only few studies report  $R_h$  fluxes during the thaw season, which strongly differ between about 32 and 240 g CO<sub>2</sub>-C m<sup>-2</sup> (Eckhardt et al., 2019; Nobrega & Grogan, 2008; Schuur et al., 2009). Heterotrophic respiration strongly depend on ecosystem conditions with highest CO<sub>2</sub> fluxes reported form moist organic tundra soils with a relatively high mean annual temperature (MAT) of  $-1^{\circ}$ C (Schuur et al., 2009) and lowest at wet, mineral tundra soils with low MAT ( $\leq$ -9°C) (Eckhardt et al., 2019; Nobrega & Grogan, 2008).

Ecosystem respiration in the arctic tundra continues during the non-growing season and may contribute a considerable fraction to the annual CO<sub>2</sub> and methane budget (Euskirchen et al., 2012; Fahnestock et al., 1998; Zimov et al., 1996; Zona et al., 2016). The two models applied in the current study behave differently at subzero temperatures. The  $Q_{10}$  model simulates CO<sub>2</sub> fluxes at any temperature below 0°C, since in situ CO, fluxes were observed even at very low winter temperatures below  $-15^{\circ}$ C (Natali et al., 2019). In contrast, the ICBM is simulating microbial CO<sub>2</sub> production only at temperatures above  $-10^{\circ}$ C. Between about -2°C and -5°C the water availability in the soil sharply decreases (Romanovsky & Osterkamp, 2000), resulting in a drop in microbial activity (Tilston et al., 2010), which becomes practically negligible below -10°C (e.g., Mikan et al., 2002; Natali et al., 2019; Schaefer & Jafarov, 2016). CO, fluxes may continue even below -10°C without microbial CO, production, e.g., by CO, diffusion along the concentration gradient between the soil and the atmosphere. The simulated CO<sub>2</sub> fluxes at sub-zero temperatures are substantial and accounted for 25%-31% of total annual fluxes (160-184 g CO<sub>2</sub>-C m<sup>-2</sup>) from heterotrophic respiration. Observations of  $R_{\rm h}$  fluxes in tundra soils during the winter season are not available but ecosystem respiration was shown to contribute a variable fraction of 15%-55% to annual CO, fluxes in Alaskan tundra soils (Euskirchen et al., 2012; Schuur et al., 2009). Since soil temperatures at the Siberian study sites are below  $-10^{\circ}$ C for five to six months of the year (Boike et al., 2019), the minor contribution of winter fluxes to total annual fluxes becomes plausible.

One of the central research questions concerning the permafrost carbon climate feedback is how fast permafrost OM will be mineralized under *in situ* conditions to  $CO_2$  and methane after thaw. However, estimates on permafrost OM decomposability based on *in situ*  $CO_2$  fluxes are still not available. Based on our calibrated ICBM, about 0.8% of thawed organic carbon is mineralized to  $CO_2$  in one year. This carbon mineralization rate is much higher than the simulated loss of 15% thawed permafrost carbon until 2100, which was derived from laboratory incubations and dynamic models (Knoblauch et al., 2013; Schuur et al., 2015). However, our field observations show a decrease of OM decomposition rates over time, which was also observed in laboratory incubations (Lee et al., 2012; Schädel et al., 2014), and is most likely caused by a decrease in the fast decomposable organic carbon pool. Hence, the initial decomposition of almost 1% of thawed permafrost organic carbon during one year will likely further slow down.

Recent upscaling of CO<sub>2</sub> flux measurements indicate that the circum-arctic permafrost region in general and the upland tundra in particular are small annual carbon sources although uncertainties are still high, particularly due to missing data during the long winter period (Belshe et al., 2013; Virkkala et al., 2021). Also multi annual CO<sub>2</sub> flux measurements on Samoylov show that the annual NEE of the polygonal tundra close to Kurungnakh fluctuates over the years (n = 8) between a carbon source and a carbon sink (+26 to  $-24 \text{ g CO}_2$ -C m<sup>-2</sup>) and it was a slight carbon source in 2016 (+7 g CO<sub>2</sub>-C m<sup>-2</sup>) (data from Holl et al., 2019). Hence, abrupt permafrost thaw, which erodes the vegetation surface and liberates old permafrost OM for



Acknowledgments The authors would like to thank the members of the joint Russian-German field campaign Lena 2016 and 2019, especially Oleg Novikov (Sukachev Institute of Forest, Krasnoyarsk), Gabriel Norén (University of Cologne), Anne Morgenstern and Guido Grosse (Alfred Wegener Institute for Polar and Marine Research, Potsdam, Germany), and Leonid Tsibizov (Trofimuk's Institute of Petroleum Geology and Geophysics, Novosibirsk, Russia), our Russian cooperation partner Dmitry Yu. Bolshiyanov (Arctic and Antarctic Research Institute, St. Petersburg, Russia) and the crew of the Russian research station Samoylov for logistical and technical support. We are grateful to Cornelia Ruhland and David Holl (Institute of Soil Science, Universität Hamburg) for help with the incubation measurements and for providing annual CO<sub>2</sub> flux data from Samoylov, respectively, and to Carolin Jarling (GFZ German Research Centre for Geosciences, Potsdam) for help with the molecular analysis of methanogens. C. Knoblauch, S. Liebner, J. Rethemeyer, and E.-M. Pfeiffer were supported by the German Ministry of Education and Research (CarboPerm Project, BMBF grant 03G0836; KoPf Project, BMBF grant 03F0764 and KOPF-Synthesis project 03F0834). C. Knoblauch and E.-M. Pfeiffer received additional support from the Clusters of Excellence CliSAP (EXC177) and CLICCS (EXC2037/1) at the Universität Hamburg funded by the German Research Foundation (DFG). CB acknowledges financial support by DFG-BE 6485/1-1. Open access funding enabled and organized by Projekt DEAL.

microbial decomposition, has a substantial effect on the annual carbon balance turning the tundra from carbon neutral to a  $CO_2$  source. However, the models used in the current study were calibrated only with  $CO_2$  flux measurements during the summer thaw period and with decomposition rates from permafrost deposits of the Lena Delta. Hence, while they improve our process understanding on *in situ*  $CO_2$  emissions from thawing permafrost substantially, the simulated  $CO_2$  fluxes are more accurate for the summer thaw period and most representative for permafrost deposits of the Lena Delta. For constraining the carbon mass balance of permafrost thawing in the northern hemisphere, pan-arctic field studies are required including the long winter period.

#### 5. Conclusions

Abrupt permafrost thaw of ice-rich Pleistocene Yedoma deposits liberates OM that is initially rapidly decomposed into mainly  $CO_2$ , turning the carbon-neutral tundra into a substantial greenhouse gas source but the depletion of the small labile carbon pool causes a decrease in  $CO_2$  fluxes over time. The formation of methane from the Pleistocene OM not only requires anaerobic conditions but also the introduction of active methanogenic communities since these are not abundant in the studied Yedoma deposits. Short-term laboratory incubations may provide an upper benchmark for *in situ*  $CO_2$  fluxes but give only very limited information on *in situ* methane fluxes. In contrast, dynamic carbon decomposition models, calibrated with long-term incubation data and  $CO_2$  field fluxes provide meaningful estimates of annual *in situ*  $CO_2$  release of which 25%–31% took place outside the summer thaw season. The bare soils are carbon sources initially after abrupt permafrost thaw, but the development of a vegetation cover will provide an additional carbon input. Therefore, multi-annual observations are required that include carbon fluxes from the upcoming vegetation and lateral fluxes of particulate and dissolved organic carbon to evaluate if these soils will stay a net carbon source, as observed during the first years after thaw, or develop to a net carbon sink in future.

# **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

#### Data Availability Statement

Data supporting the conclusions of this study are available on zenodo (https://doi.org/10.5281/ zenodo.5584710).

#### References

- Abbott, B. W., & Jones, J. B. (2015). Permafrost collapse alters soil carbon stocks, respiration, CH<sub>4</sub>, and N<sub>2</sub>O in upland tundra. *Global Change Biology*, 21(12), 4570–4587. https://doi.org/10.1111/gcb.13069
- Andrén, O., & Kätterer, T. (1997). ICBM: The introductory carbon balance model for exploration of soil carbon balances. *Ecological Applications*, 7(4), 1226–1236. https://doi.org/10.2307/264121010.1890/1051-0761(1997)007[1226:iticbm]2.0.co;2
- Beer, C. (2008). Soil science: The Arctic carbon count. Nature Geoscience, 1(9), 569-570. https://doi.org/10.1038/ngeo292
- Beer, C., Porada, P., Ekici, A., & Brakebusch, M. (2018). Effects of short-term variability of meteorological variables on soil temperature in permafrost regions. *The Cryosphere*, 12(2), 741–757. https://doi.org/10.5194/tc-12-741-2018
- Beer, C., Weber, U., Tomelleri, E., Carvalhais, N., Mahecha, M., & Reichstein, M. (2014). Harmonized European long-term climate data for assessing the effect of changing temporal variability on land–atmosphere CO<sub>2</sub> fluxes. *Journal of Climate*, 27(13), 4815–4834. https://doi. org/10.1175/jcli-d-13-00543.1
- Belshe, E. F., Schuur, E. A. G., & Bolker, B. M. (2013). Tundra ecosystems observed to be CO<sub>2</sub> sources due to differential amplification of the carbon cycle. *Ecology Letters*, 16(10), 1307–1315. https://doi.org/10.1111/ele.12164
- Boike, J., Nitzbon, J., Anders, K., Grigoriev, M., Bolshiyanov, D., Langer, M., et al. (2019). A 16-year record (2002-2017) of permafrost, active-layer, and meteorological conditions at the Samoylov Island Arctic permafrost research site, Lena River delta, northern Siberia: An opportunity to validate remote-sensing data and land surface, snow, and permafrost models. *Earth System Science Data*, 11(1), 261–299. https://doi.org/10.5194/essd-11-261-2019
- Bröder, L., Keskitalo, K., Zolkos, S., Shakil, S., Tank, S. E., Kokelj, S. V., et al. (2021). Preferential export of permafrost-derived organic matter as retrogressive thaw slumping intensifies. *Environmental Research Letters*, 16, 054059. https://doi.org/10.1088/1748-9326/abee4b Bronk Ramsey, C. (2009). Bayesian analysis of radiocarbon dates. *Radiocarbon*, 51(1), 337–360. https://doi.org/10.1017/s0033822200033865
- Cassidy, A. E., Christen, A., & Henry, G. H. R. (2016). The effect of a permafrost disturbance on growing-season carbon-dioxide fluxes in a high Arctic tundra ecosystem. *Biogeosciences*, 13(8), 2291–2303. https://doi.org/10.5194/bg-13-2291-2016
- Chadburn, S. E., Krinner, G., Porada, P., Bartsch, A., Beer, C., Belelli Marchesini, L., et al. (2017). Carbon stocks and fluxes in the high latitudes: Using site-level data to evaluate Earth system models. *Biogeosciences*, 14(22), 5143–5169. https://doi.org/10.5194/bg-14-5143-2017



- Chowdhury, R. T., Herndon, E. M., Phelps, T. J., Elias, D. A., Gu, B., Liang, L., et al. (2015). Stoichiometry and temperature sensitivity of methanogenesis and CO<sub>2</sub> production from saturated polygonal tundra in Barrow, Alaska. *Global Change Biology*, *21*(2), 722–737. https://doi.org/10.1111/gcb.12762
- Christensen, T. R., Jonasson, S., Callaghan, T. V., & Havstrom, M. (1995). Spatial variation in high latitude methane flux along a transect across Siberian and European tundra environments. *Journal of Geophysical Research - D: Atmospheres*, 100(D10), 21035–21045. https:// doi.org/10.1029/95jd02145
- Cooper, M. D. A., Estop-Aragones, C., Fisher, J. P., Thierry, A., Garnett, M. H., Charman, D. J., et al. (2017). Limited contribution of permafrost carbon to methane release from thawing peatlands. *Nature Climate Change*, 7(7), 507–511. https://doi.org/10.1038/nclimate3328
  Czudek, T., & Demek, J. (2017). Thermokarst in Siberia and its influence on the development of lowland relief. *Quaternary Research*, 1(1),
- 103–120. https://doi.org/10.1016/0033-5894(70)90013-X Davidson, S. J., Sloan, V. L., Phoenix, G. K., Wagner, R., Fisher, J. P., Oechel, W. C., & Zona, D. (2016). Vegetation type dominates the spatial variability in CH<sub>4</sub> emissions across multiple arctic tundra landscapes. *Ecosystems*, *19*(6), 1116–1132. https://doi.org/10.1007/ s10021-016-9991-0
- De Baets, S., van de Weg, M. J., Lewis, R., Steinberg, N., Meersmans, J., Quine, T. A., et al. (2016). Investigating the controls on soil organic matter decomposition in tussock tundra soil and permafrost after fire. *Soil Biology and Biochemistry*, 99, 108–116. https://doi. org/10.1016/j.soilbio.2016.04.020
- Eckhardt, T., Knoblauch, C., Kutzbach, L., Holl, D., Simpson, G., Abakumov, E., & Pfeiffer, E.-M. (2019). Partitioning net ecosystem exchange of CO<sub>2</sub> on the pedon scale in the Lena River Delta, Siberia. *Biogeosciences*, *16*(7), 1543–1562. https://doi.org/10.5194/bg-16-1543-2019
- Eckhardt, T., & Kutzbach, L. (2016). MATLAB code to calculate gas fluxes from chamber-based methods. PANGAEA. https://doi.org/10.1594/ PANGAEA.857799
- Ekici, A., Chadburn, S., Chaudhary, N., Hajdu, L. H., Marmy, A., Peng, S., et al. (2015). Site-level model intercomparison of high latitude and high altitude soil thermal dynamics in tundra and barren landscapes. *The Cryosphere*, 9(4), 1343–1361. https://doi.org/10.5194/ tc-9-1343-2015
- Elberling, B., Michelsen, A., Schädel, C., Schuur, E. A. G., Christiansen, H. H., Berg, L., et al. (2013). Long-term CO<sub>2</sub> production following permafrost thaw. *Nature Climate Change*, 3(10), 890–894. https://doi.org/10.1038/nclimate1955
- Estop-Aragonés, C., Cooper, M. D. A., Fisher, J. P., Thierry, A., Garnett, M. H., Charman, D. J., et al. (2018). Limited release of previously-frozen C and increased new peat formation after thaw in permafrost peatlands. *Soil Biology and Biochemistry*, 118, 115–129. https:// doi.org/10.1016/j.soilbio.2017.12.010
- Euskirchen, E. S., Bret-Harte, M. S., Scott, G. J., Edgar, C., & Shaver, G. R. (2012). Seasonal patterns of carbon dioxide and water fluxes in three representative tundra ecosystems in northern Alaska. *Ecosphere*, 3(1), art4. https://doi.org/10.1890/ES11-00202.1
- Fahnestock, J. T., Jones, M. H., Brooks, P. D., Walker, D. A., & Welker, J. M. (1998). Winter and early spring CO<sub>2</sub> efflux from tundra communities of northern Alaska, *Journal of Geophysical Research - D.*, 103(D22), 29023–29027, https://doi.org/10.1029/98JD00805
- Faucherre, S., Jørgensen, C. J., Blok, D., Weiss, N., Siewert, M. B., Bang-Andreasen, T. et al. (2018). Short and long-term controls on active layer and permafrost carbon turnover across the Arctic, *Journal of Geophysical Research: Biogeosciences*, 123(2), 372–390, https://doi. org/10.1002/2017JG004069
- Fouche, J., Keller, C., Allard, M., & Ambrosi, J. P. (2014). Increased CO<sub>2</sub> fluxes under warming tests and soil solution chemistry in Histic and Turbic Cryosols, Salluit, Nunavik, Canada. Soil Biology and Biochemistry, 68, 185–199. https://doi.org/10.1016/j.soilbio.2013.10.007
- Fouché, J., Keller, C., Allard, M., & Ambrosi, J. P. (2017). Diurnal evolution of the temperature sensitivity of CO<sub>2</sub> efflux in permafrost soils under control and warm conditions. *The Science of the Total Environment*, 581–582, 161–173. https://doi.org/10.1016/j. scitotenv.2016.12.089
- Gentsch, N., B. Wild, R. Mikutta, P. Čapek, K. Diáková, M. Schrumpf, et al. (2018). Temperature response of permafrost soil carbon is attenuated by mineral protection, *Global Change Biology*, 24, 3401–3415, https://doi.org/10.1111/gcb.14316
- Heimann, M., & Reichstein, M. (2008). Terrestrial ecosystem carbon dynamics and climate feedbacks. *Nature*, 451(7176), 289–292. https://doi.org/10.1038/nature06591
- Hicks Pries, C. E., Schuur, E. A. G., Natali, S. M., & Crummer, K. G. (2016). Old soil carbon losses increase with ecosystem respiration in experimentally thawed tundra. *Nature Climate Change*, 6(2), 214–218. https://doi.org/10.1038/nclimate2830
- Høj, L., Olsen, R. A., & Torsvik, V. L. (2007). Effects of temperature on the diversity and community structure of known methanogenic groups and other archaea in high Arctic peat. *The ISME Journal*, 2, 37–48. https://doi.org/10.1038/ismej.2007.84
- Holl, D., Wille, C., Sachs, T., Schreiber, P., Runkle, B. R. K., Beckebanze, L., et al. (2019). A long-term (2002 to 2017) record of closed-path and open-path eddy covariance CO<sub>2</sub> net ecosystem exchange fluxes from the Siberian Arctic. *Earth System Science Data*, 11(1), 221–240. https://doi.org/10.5194/essd-11-221-2019
- Holm, S., Walz, J., Horn, F., Yang, S., Grigoriev, M. N., Wagner, D., et al. (2020). Methanogenic response to long-term permafrost thaw is determined by paleoenvironment. *FEMS Microbiology Ecology*, 96(3). https://doi.org/10.1093/femsec/fiaa021
- Hugelius, G., Strauss, J., Zubrzycki, S., Harden, J. W., Schuur, E. A. G., Ping, C.-L., et al. (2014). Estimated stocks of circumpolar permafrost carbon with quantified uncertainty ranges and identified data gaps. *Biogeosciences*, 11(23), 6573–6593. https://doi.org/10.5194/ bg-11-6573-2014
- Jensen, A. E., Lohse, K. A., Crosby, B. T., & Mora, C. I. (2014). Variations in soil carbon dioxide efflux across a thaw slump chronosequence in northwestern Alaska. Environmental Research Letters, 9(2), 025001. https://doi.org/10.1088/1748-9326/9/2/025001

Jørgensen, C. J., Johansen, K. M. L., Westergaard-Nielsen, A., & Elberling, B. (2015). Net regional methane sink in High Arctic soils of northeast Greenland. Nature Geoscience, 8(1), 20–23. https://doi.org/10.1038/ngeo2305

- Keiluweit, M., Nico, P. S., Kleber, M., & Fendorf, S. (2016). Are oxygen limitations under recognized regulators of organic carbon turnover in upland soils? *Biogeochemistry*, 127(2–3), 157–171. https://doi.org/10.1007/s10533-015-0180-6
- Knoblauch, C., Beer, C., Liebner, S., Grigoriev, M. N., & Pfeiffer, E.-M. (2018). Methane production as key to the greenhouse gas budget of thawing permafrost. *Nature Climate Change*, 8(4), 309–312. https://doi.org/10.1038/s41558-018-0095-z
- Knoblauch, C., Beer, C., Sosnin, A., Wagner, D., & Pfeiffer, E.-M. (2013). Predicting long-term carbon mineralization and trace gas production from thawing permafrost of Northeast Siberia. *Global Change Biology*, 19(4), 1160–1172. https://doi.org/10.1111/gcb.12116
- Knoblauch, C., Zimmermann, U., Blumenberg, M., Michaelis, W., & Pfeiffer, E.-M. (2008). Methane turnover and temperature response of methane-oxidising bacteria in permafrost-affected soils of northeast Siberia. Soil Biology and Biochemistry, 40, 3004–3013. https://doi.org/10.1016/j.soilbio.2008.08.020



- Koven, C. D., Schuur, E. A. G., Schädel, C., Bohn, T. J., Burke, E. J., Chen, G., et al. (2015). A simplified, data-constrained approach to estimate the permafrost carbon–climate feedback. *Philosophical Transactions of the Royal. Society A*, 373(2054), https://doi.org/10.1098/ rsta.2014.0423
- Kutzbach, L., Wagner, D., & Pfeiffer, E.-M. (2004). Effect of microrelief and vegetation on methane emission from wet polygonal tundra, Lena Delta, Northern Siberia. *Biogeochemistry*, 69, 341–362. https://doi.org/10.1023/b:biog.0000031053.81520.db
- Kwon, M. J., Beulig, F., Ilie, I., Wildner, M., Küsel, K., Merbold, L., et al. (2017). Plants, microorganisms, and soil temperatures contribute to a decrease in methane fluxes on a drained Arctic floodplain. *Global Change Biology*, 23(6), 2396–2412. https://doi.org/10.1111/gcb.13558 Lee, H., Schuur, E. A. G., Inglett, K. S., Lavoie, M., & Chanton, J. P. (2012). The rate of permafrost carbon release under aerobic and anaerobic
- Liebner, S., Ganzert, L., Kiss, A., Yang, S. Z., Wagner, D., & Svenning, M. M. (2015). Shifts in methanogenic community composition
- and methane fluxes along the degradation of discontinuous permafrost. *Frontiers in Microbiology*, *6*, 356. https://doi.org/10.3389/ fmicb.2015.00356
- MacDougall, A. H., & Knutti, R. (2016). Projecting the release of carbon from permafrost soils using a perturbed parameter ensemble modelling approach. *Biogeosciences*, 13(7), 2123–2136. https://doi.org/10.5194/bg-13-2123-2016
- Mauritz, M., Bracho, R., Celis, G., Hutchings, J., Natali, S. M., Pegoraro, E., et al. (2017). Nonlinear CO<sub>2</sub> flux response to 7 years of experimentally induced permafrost thaw. *Global Change Biology*, 23(9), 3646–3666. https://doi.org/10.1111/gcb.13661
- Mauritz, M., Pegoraro, E., Ogle, K., Ebert, C., & Schuur, E. A. G. (2021). Investigating thaw and plant productivity constraints on old soil carbon respiration from permafrost. *Biogeosciences*, 126(6), e2020JG006000. https://doi.org/10.1029/2020JG006000
- McGuire, A. D., Lawrence, D. M., Koven, C., Clein, J. S., Burke, E., Chen, G., et al. (2018). Dependence of the evolution of carbon dynamics in the northern permafrost region on the trajectory of climate change. *Proceedings of the National Academy of Sciences of the United States of America*, 115(15), 3882–3887. https://doi.org/10.1073/pnas.1719903115
- Mikan, C. J., Schimel, J. P., & Doyle, A. P. (2002). Temperature controls of microbial respiration in arctic tundra soils above and below freezing. Soil Biology and Biochemistry, 34(11), 1785–1795. https://doi.org/10.1016/s0038-0717(02)00168-2
- Moni, C., Lerch, T. Z., Knoth de Zarruk, K., Strand, L. T., Forte, C., Certini, G., & Rasse, D. P. (2015). Temperature response of soil organic matter mineralisation in arctic soil profiles. Soil Biology and Biochemistry, 88, 236–246. https://doi.org/10.1016/j.soilbio.2015.05.024
- Monteux, S., Keuper, F., Fontaine, S., Gavazov, K., Hallin, S., Juhanson, J., et al. (2020). Carbon and nitrogen cycling in Yedoma permafrost controlled by microbial functional limitations. *Nature Geoscience*, 13(12), 794–798. https://doi.org/10.1038/s41561-020-00662-4
- Mueller, C. W., Rethemeyer, J., Kao-Kniffin, J., Löppmann, S., Hinkel, K. M., & Bockheim, J. G. (2015). Large amounts of labile organic carbon in permafrost soils of northern Alaska. *Global Change Biology*, 21(7), 2804–2817. https://doi.org/10.1111/gcb.12876
- Natali, S. M., Schuur, E. A. G., Mauritz, M., Schade, J. D., Celis, G., Crummer, K. G., et al. (2015). Permafrost thaw and soil moisture driving CO<sub>2</sub> and CH<sub>4</sub> release from upland tundra. *Journal of Geophysical Research: Biogeosciences*, 120(3), 525–537. https://doi. org/10.1002/2014jg002872
- Natali, S. M., Watts, J. D., Rogers, B. M., Potter, S., Ludwig, S. M., Selbmann, A.-K., et al. (2019). Large loss of CO<sub>2</sub> in winter observed across the northern permafrost region. *Nature Climate Change*, 9(11), 852–857. https://doi.org/10.1038/s41558-019-0592-8
- Nobrega, S., & Grogan, P. (2008). Landscape and ecosystem-level controls on net carbon dioxide exchange along a natural moisture gradient in Canadian low arctic tundra. *Ecosystems*, 11(3), 377–396. https://doi.org/10.1007/s10021-008-9128-1
- Olefeldt, D., Turetsky, M. R., Crill, P. M., & McGuire, A. D. (2013). Environmental and physical controls on northern terrestrial methane emissions across permafrost zones. *Global Change Biology*, 19(2), 589–603. https://doi.org/10.1111/gcb.12071
- Pegoraro, E., Mauritz, M., Bracho, R., Ebert, C., Dijkstra, P., Hungate, B. A., et al. (2019). Glucose addition increases the magnitude and decreases the age of soil respired carbon in a long-term permafrost incubation study. *Soil Biology and Biochemistry*, *129*, 201–211. https://doi.org/10.1016/j.soilbio.2018.10.009
- Pegoraro, E. F., Mauritz, M. E., Ogle, K., Ebert, C. H., & Schuur, E. A. G. (2021). Lower soil moisture and deep soil temperatures in thermokarst features increase old soil carbon loss after 10 years of experimental permafrost warming. *Global Change Biology*, 27(6), 1293–1308. https://doi.org/10.1111/gcb.15481
- Plaza, C., Pegoraro, E., Bracho, R., Celis, G., Crummer, K. G., Hutchings, J. A., et al. (2019). Direct observation of permafrost degradation and rapid soil carbon loss in tundra. *Nature Geoscience*, 12(8), 627–631. https://doi.org/10.1038/s41561-019-0387-6
- Porada, P., Lenton, T. M., Pohl, A., Weber, B., Mander, L., Donnadieu, Y., et al. (2016). High potential for weathering and climate effects of non-vascular vegetation in the Late Ordovician. *Nature Communications*, 7, 12113. https://doi.org/10.1038/ncomms12113
- Preuss, I., Knoblauch, C., Gebert, J., & Pfeiffer, E.-M. (2013). Improved quantification of microbial CH<sub>4</sub> oxidation efficiency in arctic wetland soils using carbon isotope fractionation. *Biogeosciences*, 10(4), 2539–2552. https://doi.org/10.5194/bg-10-2539-2013
- Ravn, N. R., Elberling, B., & Michelsen, A. (2020). Arctic soil carbon turnover controlled by experimental snow addition, summer warming and shrub removal. Soil Biology and Biochemistry, 142, 107698. https://doi.org/10.1016/j.soilbio.2019.107698
- Romanovsky, V. E., & Osterkamp, T. E. (2000). Effects of unfrozen water on heat and mass transport processes in the active layer and permafrost. *Permafrost Periglac*, 11(3), 219–239. https://doi.org/10.1002/1099-1530(200007/09)11:3<219::aid-ppp352>3.0.co;2-7
- Schädel, C., Bader, M. K. F., Schuur, E. A. G., Biasi, C., Bracho, R., Capek, P., et al. (2016). Potential carbon emissions dominated by carbon dioxide from thawed permafrost soils. *Nature Climate Change*, 6, 950–953. https://doi.org/10.1038/nclimate3054
- Schädel, C., Schuur, E. A. G., Bracho, R., Elberling, B., Knoblauch, C., Lee, H., et al. (2014). Circumpolar assessment of permafrost C quality and its vulnerability over time using long-term incubation data. *Global Change Biology*, 20(2), 641–652. https://doi.org/10.1111/ gcb.12417
- Schaefer, K., & Jafarov, E. (2016). A parameterization of respiration in frozen soils based on substrate availability. *Biogeosciences*, 13(7), 1991–2001. https://doi.org/10.5194/bg-13-1991-2016
- Schink, B. (1997). Energetics of syntrophic cooperation in methanogenic degradation. Microbiology and Molecular Biology Reviews, 61(2), 262–280. https://doi.org/10.1128/mmbr.61.2.262-280.1997
- Schneider von Deimling, T., Grosse, G., Strauss, J., Schirrmeister, L., Morgenstern, A., Schaphoff, S., et al. (2015). Observation-based modelling of permafrost carbon fluxes with accounting for deep carbon deposits and thermokarst activity. *Biogeosciences*, *12*(11), 3469–3488. https://doi.org/10.5194/bg-12-3469-2015
- Schuur, E. A. G., McGuire, A. D., Schädel, C., Grosse, G., Harden, J. W., Hayes, D. J., et al. (2015). Climate change and the permafrost carbon feedback. *Nature*, 520(7546), 171–179. https://doi.org/10.1038/nature14338
- Schuur, E. A. G., Vogel, J. G., Crummer, K. G., Lee, H., Sickman, J. O., & Osterkamp, T. E. (2009). The effect of permafrost thaw on old carbon release and net carbon exchange from tundra. *Nature*, 459(7246), 556–559. https://doi.org/10.1038/nature08031
- Sexstone, A. J., Parkin, T. B., & Tiedje, J. M. (1988). Denitrification response to soil wetting in aggregated and unaggregated soil. Soil Biology and Biochemistry, 20(5), 767–769. https://doi.org/10.1016/0038-0717(88)90165-4



- Song, X. Y., Wang, G. X., Ran, F., Huang, K. W., Sun, J. Y., & Song, C. L. (2020). Soil moisture as a key factor in carbon release from thawing permafrost in a boreal forest. *Geoderma*, 357, 9. https://doi.org/10.1016/j.geoderma.2019.113975
- Steinhof, A., Altenburg, M., & Machts, H. (2017). Sample preparation at the Jena <sup>14</sup>C laboratory. *Radiocarbon*, 59(3), 815–830. https://doi. org/10.1017/RDC.2017.50
- Strauss, J., Schirrmeister, L., Mangelsdorf, K., Eichhorn, L., Wetterich, S., & Herzschuh, U. (2015). Organic-matter quality of deep permafrost carbon—A study from Arctic Siberia. *Biogeosciences*, 12(7), 2227–2245. https://doi.org/10.5194/bg-12-2227-2015
- Ström, L., Falk, J. M., Skov, K., Jackowicz-Korczynski, M., Mastepanov, M., Christensen, T. R., et al. (2015). Controls of spatial and temporal variability in CH<sub>4</sub> flux in a high arctic fen over three years. *Biogeochemistry*, 125(1), 21–35. https://doi.org/10.1007/s10533-015-0109-0
- Tanski, G., Lantuit, H., Ruttor, S., Knoblauch, C., Radosavljevic, B., Strauss, J., et al. (2017). Transformation of terrestrial organic matter along thermokarst-affected permafrost coasts in the Arctic. *The Science of the Total Environment*, 581–582, 434–447. https://doi. org/10.1016/j.scitotenv.2016.12.152
- Tilston, E. L., Sparrman, T., & Öquist, M. G. (2010). Unfrozen water content moderates temperature dependence of sub-zero microbial respiration. *Soil Biology and Biochemistry*, 42(9), 1396–1407. https://doi.org/10.1016/j.soilbio.2010.04.018
- Turetsky, M. R., Abbott, B. W., Jones, M. C., Anthony, K. W., Olefeldt, D., Schuur, E. A. G., et al. (2020). Carbon release through abrupt permafrost thaw. *Nature Geoscience*, 13(2), 138–143. https://doi.org/10.1038/s41561-019-0526-0
- van Hulzen, J. B., Segers, R., van Bodegom, P. M., Leffelaar, P. A., Schmid, M., De Batist, M., et al. (1999). Temperature effects on soil methane production: An explanation for observed variability. Soil Biology and Biochemistry, 31(14), 1919–1929. https://doi.org/10.1016/ s0038-0717(99)00109-1
- Vaughn, L. J. S., Conrad, M. E., Bill, M., & Torn, M. S. (2016). Isotopic insights into methane production, oxidation, and emissions in Arctic polygon tundra. *Global Change Biology*, 22(10), 3487–3502. https://doi.org/10.1111/gcb.13281
- Vaughn, L. J. S., & Torn, M. S. (2019). <sup>14</sup>C evidence that millennial and fast-cycling soil carbon are equally sensitive to warming. Nature Climate Change, 9(6), 467–471. https://doi.org/10.1038/s41558-019-0468-y
- Viovy, N. (2018). CRUNCEP version 7 atmospheric forcing data for the community land model. Research Data Archive at the National Center for Atmospheric Research, Computational and Information Systems Laboratory.
- Virkkala, A.-M., Aalto, J., Rogers, B. M., Tagesson, T., Treat, C. C., Natali, S. M., et al. (2021). Statistical upscaling of ecosystem CO<sub>2</sub> fluxes across the terrestrial tundra and boreal domain: Regional patterns and uncertainties. *Global Change Biology*. https://doi.org/10.1111/ gcb.15659
- von Fischer, J. C., & Hedin, L. O. (2002). Separating methane production and consumption with a field-based isotope pool dilution technique. *Global Biogeochemical Cycles*, 16(3), 1034. https://doi.org/10.1029/2001gb001448
- Vonk, J. E., Sanchez-Garcia, L., van Dongen, B. E., Alling, V., Kosmach, D., Charkin, A., et al. (2012). Activation of old carbon by erosion of coastal and subsea permafrost in Arctic Siberia. *Nature*, 489(7414), 137–140. https://doi.org/10.1038/nature11392
- Wagner, D. (2017). Effect of varying soil water potentials on methanogenesis in aerated marshland soils. *Scientific Reports*, 7, 14706. https://doi.org/10.1038/s41598-017-14980-y
- Waldrop, M. P., Wickland, K. P., White, R., III, Berhe, A. A., Harden, J. W., & Romanovsky, V. E. (2010). Molecular investigations into a globally important carbon pool: Permafrost-protected carbon in Alaskan soils. *Global Change Biology*, 16(9), 2543–2554. https://doi. org/10.1111/j.1365-2486.2009.02141.x
- Walz, J., Knoblauch, C., Böhme, L., & Pfeiffer, E.-M. (2017). Regulation of soil organic matter decomposition in permafrost-affected Siberian tundra soils—Impact of oxygen availability, freezing and thawing, temperature, and labile organic matter. Soil Biology and Biochemistry, 110, 34–43. https://doi.org/10.1016/j.soilbio.2017.03.001
- Walz, J., Knoblauch, C., Tigges, R., Opel, T., Schirrmeister, L., & Pfeiffer, E.-M. (2018). Greenhouse gas production in degrading ice-rich permafrost deposits in northeastern Siberia. *Biogeosciences*, 15(17), 5423–5436. https://doi.org/10.5194/bg-15-5423-2018
- Wang, J., Wu, Q., Yuan, Z., & Kang, H. (2020). Soil respiration of alpine meadow is controlled by freeze-thaw processes of active layer in the permafrost region of the Qinghai–Tibet Plateau. *The Cryosphere*, 14(9), 2835–2848. https://doi.org/10.5194/tc-14-2835-2020
- Wang, K. F., Peng, C. H., Zhu, Q. A., Zhou, X. L., Wang, M., Zhang, K. R., & Wang, G. S. (2017). Modeling global soil carbon and soil microbial carbon by integrating microbial processes into the ecosystem process model TRIPLEX-GHG. *Journal of Advances in Modeling Earth Systems*, 9(6), 2368–2384. https://doi.org/10.1002/2017ms000920
- Wetterich, S., Kuzmina, S., Andreev, A. A., Kienast, F., Meyer, H., Schirrmeister, L., et al. (2008). Palaeoenvironmental dynamics inferred from late quaternary permafrost deposits on Kurungnakh Island, Lena Delta, Northeast Siberia, Russia. *Quaternary Science Reviews*, 27(15–16), 1523–1540. https://doi.org/10.1016/j.quascirev.2008.04.007
- Wild, B., Gentsch, N., Čapek, P., Diáková, K., Alves, R. J. E., Bárta, J., et al. (2016). Plant-derived compounds stimulate the decomposition of organic matter in arctic permafrost soils. *Scientific Reports*, 6, 25607. https://doi.org/10.1038/srep25607
- WRB, I. W. g. (2014). World reference base for soil Resources 2014 international soil classification system for naming soils and creating legends for soil maps. FAO.
- Zheng, J., RoyChowdhury, T., Yang, Z., Gu, B., Wullschleger, S. D., & Graham, D. E. (2018). Impacts of temperature and soil characteristics on methane production and oxidation in Arctic tundra. *Biogeosciences*, 15(21), 6621–6635. https://doi.org/10.5194/bg-15-6621-2018
- Zimov, S. A., Davidov, S. P., Voropaev, Y. V., Prosiannikov, S. F., Semiletov, I. P., Chapin, M. C., & Chapin, F. S. (1996). Siberian CO<sub>2</sub> efflux in winter as a CO<sub>2</sub> source and cause of seasonality in atmospheric CO<sub>2</sub>. *Climatic Change*, 33(1), 111–120. https://doi.org/10.1007/bf00140516
- Zimov, S. A., Davydov, S. P., Zimova, G. M., Davydova, A. I., Schuur, E. A. G., Dutta, K., & Chapin, F. S. (2006). Permafrost carbon: Stock and decomposability of a globally significant carbon pool. *Geophysical Research Letters*, 33(20), L20502. https://doi.org/10.1029/2006gl027484
- Zona, D., Gioli, B., Commane, R., Lindaas, J., Wofsy, S. C., Miller, C. E., et al. (2016). Cold season emissions dominate the Arctic tundra methane budget. Proceedings of the National Academy of Sciences of the United States of America, 113(1), 40–45. https://doi.org/10.1073/ pnas.1516017113